Author's response to reviews

Title: A randomised controlled trial investigating the effects of Mediterranean diet and aerobic exercise on cognition in cognitively healthy older people living independently within aged care facilities: the Lifestyle Intervention in Independent Living Aged Care (LIILAC) study protocol

Authors:
Roy J Hardman (rhardman@swin.edu.au)
Greg Kennedy (gikennedy@swin.edu.au)
Helen MacPherson (helen.macpherson@deakin.edu.au)
Andrew B Scholey (ascholey@swin.edu.au)
Andrew Pipingas (apipingas@swin.edu.au)

Version: 4 Date: 1 May 2015

Author's response to reviews:

Dear Editor,

The updated version of our manuscript submitted earlier today included and outdated version of our flow diagram. Please accept our updated version with the correct diagram attached.

There have been no other alterations to the manuscript.

Please accept our apologies for the inconvenience.

Kind Regards,

Greg Kennedy